Author's response to reviews

Title: OPAL: Randomized, controlled clinical trial to investigate oxygen persufflation as adjunct in liver preservation

Authors:

Thomas Minor (tminor@uni-bonn.de)
Carolin Pütter (Carolin.puetter@uk-essen.de)
Anja Gallinat (a.gallinat@gmx.de)
Claudia Ose (Claudia.Ose@uk-essen.de)
Gernot Kaiser (Gernot.Kaiser@uk-essen.de)
Andre Scherag (Andre.Scherag@uk-essen.de)
Jürgen Treckmann (j.treckmann@gmx.net)
Andreas Paul (Andreas.Paul@uk-essen.de)

Version: 4 Date: 26 October 2011

Author's response to reviews: see over
Dear Sir!

Please find enclosed the re-formatted manuscript of our study protocol entitled:

' **Oxygen persufflation as adjunct in liver preservation (OPAL): study protocol for a randomized controlled trial**, '

submitted for publication in Trials.

We have changed the title according to the journal style for study protocol articles as outlined by the associate editor.

In addition we have removed to upload of the documents certifying ethics approval and financial funding.

We hope that the reformatted manuscript will now be acceptable for publication in your esteemed journal and we are looking forward to hearing from you in the near future.

With best regards
Yours,

T. Minor